Author's response to reviews

Title: TIPIT: A Randomised Controlled Trial of Thyroxine in Preterm Infants under 28 weeks Gestation: Magnetic Resonance Imaging and Magnetic Resonance Angiography Protocol

Authors:

Sze M Ng (M.Ng@liverpool.ac.uk)
Mark A Turner (mark.turner@liverpool.ac.uk)
Carrol Gamble (C.gamble@liverpool.ac.uk)
Mohammed Didi (mohammed.didi@rlc.nhs.uk)
Suresh Victor (Suresh.victor@manchester.ac.uk)
Christina Malamateniou (Christina.Malamateniou@liverpool.ac.uk)
Laura Parkes (laupar@liverpool.ac.uk)
Anna Tietze (anna.tietze@rlc.nhs.uk)
Lloyd Gregory (lloyd.gregory@manchester.ac.uk)
Vanessa Sluming (vanessa.sluming@liverpool.ac.uk)
Laurence Abernethy (laurence.abernethy@rlc.nhs.uk)
Alan Michael Weindling (a.m.weindling@liverpool.ac.uk)

Version: 2 Date: 17 April 2008

Author's response to reviews:

We have initially submitted the manuscript to BMC Medicine. Unfortunately, BMC Medicine does not consider study protocols for publication.

We would like to submit this study protocol to BMC Pediatrics. The protocol describes an RCT studying the effect of thyroxine on the extremely premature brain at 40 weeks gestation using Magnetic Resonance imaging and Magnetic resonance angiography.